Author's response to reviews

Title: Unsuspected Pneumocystis pneumonia in an HIV-seronegative patient with untreated lung cancer: a case report

Authors:

Cai Chuang (skinblack1966@yahoo.com.cn)
Xie Zhanhong (Drxiezhanhong@163.com)
Gu Yinyin (DrGuyinyin@163.com)
Zhong Shuqing (Drzhongshuqing@163.com)
Zhong Nanshan (DrNanshanzhong@163.com)

Version: 6 Date: 25 September 2007

Author's response to reviews:

Dear Editor,

Thank you very much for the good news. Please inform us in detail how to pay the fee for the publication of the article once it has been fully accepted.

Because the patient has died, her husband, the next kin of the patient, was asked to sign the patient consent form, and he had kindly complied.

So we as the authors of this manuscript can declare that "Written informed consent was obtained from the patient's next of kin(husband) for publication of this case report and any accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal."

Best wishes

Caichuang, corresponding author of the manuscript.

2007-9-25